Author's response to reviews

Title: Bilateral Thoracic Paravertebral Block for Immediate Postoperative Pain Relief in the PACU: A Prospective, Observational Study

Authors:

fei liu (30333870@qq.com)
Huan-Kai Zhang (zhanghuankaimazui@163.com)
Yun-Xia Zuo (769574913@qq.com)

Version: 2 Date: 08 Jun 2017

Author’s response to reviews:

Dear Guangde Tu:

Re: Manuscript ID BANE-D-16-00364, entitled “Bilateral Thoracic Paravertebral Block for Immediate Postoperative Pain Relief in the PACU: A Prospective, Observational Study”

Thank you for offering us this opportunity to amend our manuscript. We have followed your suggestion to add the trial registration number in the last line of the abstract of the manuscript.

Yours sincerely

Yunxia Zuo
on behalf of all authors